Author's response to reviews

Title: Two novel connexin32 mutations cause early onset X-linked Charcot-Marie-Tooth

Authors:

Geir J Braathen (g.j.braathen@medisin.uio.no)
Jette C Sand (jette.sand@ahus.no)
Geir Bukholm (geir.bukholm@medisin.uio.no)
Michael B Russell (m.b.russell@medisin.uio.no)

Version: 2 Date: 2 May 2007

Author's response to reviews:

Dear editor

We have cautiously edited the manuscript according to the reviewer recommendations. Please find enclosed our reply to the reviewers comments. We prefer publication in BMC Neurology.

Yours sincerely

Geir J. Braathen
MD, neurologist and geneticist
University of Oslo